Author's response to reviews

Title: Protocol for the Locomotor Experience Applied Post-stroke (LEAPS) trial: A randomized controlled trial

Authors:

Pamela W Duncan (Pamela.Duncan@Duke.edu)
Katherine J Sullivan (kasulliv@usc.edu)
Andrea L Behrman (abehrman@phhp.ufl.edu)
Stanley P Azen (sazen@usc.edu)
Samuel S. Wu (samwu@biostat.ufl.edu)
Stephen E Nadeau (snadeau@ufl.edu)
Bruce H Dobkin (bdobkin@mednet.ucla.edu)
Dorian K Rose (drose@aging.ufl.edu)
Julie K Tilson (tilson@usc.edu)

Version: 4 Date: 10 September 2007

Author's response to reviews: see over
September 6, 2007

BMC Neurology
BioMed Central
Middlesex House
32-42 Cleveland Street
London
W1T4LB
UK

To the Editor:

Thank you for your careful review of this manuscript. The requested changes have been made and are outlined below.

1. Ethics and consent: The photographs to be published with this manuscript contain images of LEAPS research participants. The signed consents to use these images will be faxed to the editorial office.
2. The manuscript has been revised to detail the information regarding ethics committee approvals at all sites. All research conducted in association with this study is in compliance with the Helsinki declaration and with local, state and federal guidelines at all sites. Informed consent has been obtained and documented from all individuals participating in this study.
3. Author’s contributions: This section has been revised.
4. Abstract: The abstract has been revised such that it is now structured according to the guidelines provided.
5. The revised manuscript has been reviewed for conformity with the journal style requested.

Thank you for your consideration of this manuscript.

Sincerely,

Pamela W. Duncan, Ph.D., FAPTA
Professor and Bette Busch Maniscalco Research Fellow
Doctor of Physical Therapy Division
Department of Community and Family Medicine
Senior Fellow; Duke Center for Clinical Health Policy Research